Author's response to reviews

**Title:** Splinting after contracture release for Dupuytren's contracture (SCoRD): protocol of a pragmatic, multi-centre, randomized controlled trial

**Authors:**

Christina Jerosch-Herold (c.jerosch-herold@uea.ac.uk)
Lee Shepstone (L.shepstone@uea.ac.uk)
Adrian J Chojnowski (adrian.chojnowski@nnuh.nhs.uk)
Debbie Larson (Debbie.larson@nnuh.nhs.uk)

**Version:** 2 **Date:** 25 April 2008

**Author's response to reviews:**

Formatting change as requested have been made:

1. author list now corresponds to the one in the submission system

2. a conclusion has not been provided as this is a trial protocol and therefore it is impossible to draw any conclusions

3. all references have been checked and the number in the list corresponds to the number extracted in the submission system